Author's response to reviews

Title: Update on the NCEP ATP-III Emerging Cardiometabolic Risk Factors

Authors:

Robert H Eckel (robert.eckel@ucdenver.edu)
Marc-Andre Cornier (marc.cornier@ucdenver.edu)

Version: 3 Date: 19 May 2014

Author's response to reviews: see over
May 18, 2014

RE: MS#2120362812449199 – Update on the NCEP ATP-III Emerging Cariometabolic Risk Factors

Ursula D’Souza, PhD
Senior Editor
BMC Medicine

Dear Dr. D’Souza,

We would like to thank you for considering and “re-reviewing” our manuscript titled “Update on the NCEP ATP-III Emerging Cariometabolic Risk Factors”. We have added the three references as recommended by Reviewer 2 (Lloyd-Jones Circ 2010;121:1768; Tzoulaki et al JAMA 2009;302:2345; van den Oord et al Atherosclerosis 2013;228:1-11).

Thank you again for reconsidering this manuscript. If you have any questions please do not hesitate to contact me at (303) 724-3923 or at robert.eckel@ucdenver.edu.

Sincerely,

Robert H. Eckel, M.D.
Professor of Medicine
   Division of Endocrinology, Metabolism and Diabetes
   Division of Cardiology
Professor of Physiology and Biophysics
Charles A. Boettcher II Chair in Atherosclerosis
Director Discovery Translation, Colorado Clinical and Translational Science Institute
University of Colorado Denver School of Medicine
Director Lipid Clinic, University Hospital